Author's response to reviews

Title: An evaluation of POSSUM and P-POSSUM scoring in predicting post-operative mortality in a level 1 critical care setting

Authors:

Sarah Scott (sarah.scott29@nhs.net)
Jonathan N Lund (jon.lund@nottingham.ac.uk)
Stuart Gold (stuart.gold@nhs.net)
Richard Elliot (richard.elliott3@nhs.net)
Mair Vater (mair.vater@nhs.net)
Mallicka Chakrabarty (dr_mallicka@yahoo.com)
Thomas P Heinink (tomheinink@doctors.org.uk)
John P Williams (john.williams7@nhs.net)

Version: 3 Date: 23 September 2014

Author's response to reviews: see over
23rd September 2014

The Editor
BMC Anesthesiology
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom

Editor

An evaluation of POSSUM and P-POSSUM scoring in predicting post-operative mortality in a level 1 critical care setting.

Thank you for your comments on our revisions to the above manuscript, and suggested changes. Please find enclosed a manuscript incorporating these changes. The column headings in table 4 have been changed as requested. Author email addresses are included, and the contributions of each author have been clarified.

Sincerely,

Dr Thomas P Heinink BMedSci, BMBS, FRCA
Clinical Research Fellow in Anaesthesia
Department of Anaesthesia & Intensive Care
Derby Hospitals NHS Foundation Trust, Royal Derby Hospital
Uttoxeter Road
Derby
DE22 3NE
United Kingdom
tomheinink@doctors.org.uk